Author’s response to reviews

Title: Meta-analysis on resected pancreatic cancer: A comparison between adjuvant treatments and gemcitabine alone

Authors:

HuaChen HuaChen (1441530979@qq.com)
Ruizhi He (heruizhi1020@qq.com)
Xiuhui Shi (shi-xiuhui@qq.com)
Min Zhou (1550230569@qq.com)
Chunle Zhao (dr.zhaocl@outlook.com)
Hang Zhang (172356995@qq.com)
Renyi Qin (ryqin@tjh.tjmu.edu.cn)

Version: 1 Date: 27 Sep 2018

Author’s response to reviews:

The reviewer’s comments have been answered and explained one by one in the attachment.